Author’s response to reviews

Title: Associations Between Open Drain Flooding and Pediatric Enteric Infections in the MAL-ED Cohort in a Low-Income, Urban Neighborhood in Vellore, India

Authors:

David Berendes (dberendes@cdc.gov)
Juan Leon (juan.leon@emory.edu)
Amy Kirby (agkl@cdc.gov)
Julie Clennon (jclenno@emory.edu)
Suraja Raj (suraja.jeya.raj@emory.edu)
Habib Yakubu (hyakubu@emory.edu)
Katharine Robb (karobb@emory.edu)
Arun Kartikeyan (dr.arunks@gmail.com)
Priya Hemavathy (priya.hemavathy82@gmail.com)
Annai Gunasekaran (annaimbavellore@gmail.com)
Sheela Roy (sheelaroy1@gmail.com)
Ben Ghale (benowr@live.com)
J Senthil Kumar (senthilcmc2008@gmail.com)
Venkata Mohan (venkat@cmcvellore.ac.in)
Gagandeep Kang (gkang@cmcvellore.ac.in)
Christine Moe (clmoe@emory.edu)

Version: 4 Date: 26 Jun 2019

Author’s response to reviews:
Many thanks for the review. We have made the copyright changes to the Figure caption in the manuscript (requested change #1). For the English editing (request #2), an assistant BMC PH editor (Rosie Brooks) reviewed the manuscript and could not locate any textual errors that would warrant further copy edit (email on June 25, 2019), thus we disregarded this comment.